Author's response to reviews

Title: Motorbike Rider With Ventricular Arrhythmia - Suspected Association to Air Pollution

Authors:

    Kent Emilsson (kent.emilsson@orebroll.se)

Version: 2 Date: 6 December 2007

Author's response to reviews: see over
Re: Manuscript entitled: “Moped rider with ventricular arrhythmia - suspect association with air pollution.”

Dear Editor-in-Chief,

The manuscript with the above mentioned title, which now has been changed to ”Motorbike Rider With Ventricular Arrhythmia - Suspected Association to Air Pollution”, has now been checked over by a native English speaker and revised. I believe that this manuscript provides important information regarding the association between air pollution and ventricular arrhythmia. The manuscript has not been previously published and is not under simultaneous consideration by any other publication. There is no personal conflict of interest.

I thank you and your reviewers in advance for the time taken with this manuscript.

Yours sincerely

Kent Emilsson, MD, PhD
Department of Clinical Physiology
Örebro University Hospital
SE-701 85 ÖREBRO
SWEDEN
Telephone number: +460196025978
E-mail: kent.emilsson@orebroll.se (preferred)

Attached
Revised manuscript